Author’s response to reviews

Title: Association between pregnancy intention and optimal breastfeeding practices in the Philippines: a cross-sectional study

Authors:

Valerie Gilbert T. Ulep (vgtulep@gmail.com)
Maridel P Borja (dedelborja@yahoo.com)

Version: 5 Date: 2 July 2012

Author’s response to reviews: see over
July 2, 2012

Dear Rachel Neilan,

I would like to express my gratitude in peer reviewing my manuscript. In response to your comments, attached herewith are my answers to your comments/questions.

I hope you find everything in order.

Sincerely,

Valerie Gilbert Ulep
Please clarify the ethical approval and/or informed consent given for your work. Full details of this (including name of ethics committee and reference number if applicable) should be contained within the methods section of your manuscript.

- Please see page 4 (in red highlight)